Author’s response to reviews

Title: Development of spasticity with age in a total population of children with cerebral palsy

Authors:

Gunnar Hagglund (gunnar.hagglund@med.lu.se)
Philippe Wagner (PW@nko.se)

Version: 6 Date: 1 November 2008

Author’s response to reviews:

Abstract:
Heading "Methods" should be in bold letters
Background
Para 2 "During the first year of life..." Change to "During the first years of life..
Methods
Para 5
"...and +1" Change to "...nad 1+
Reference 1 "c" in Scholtes written in read
Table 1
Delete "ref)". New spelling: Table 1. The modified Ashworth scale
Table 3
Different space between lines in the table.

Yours sincerely
Gunnar Hägglund